Author's response to reviews

Title: Removal of urothelium affects bladder contractility and release of ATP but not release of NO in rat urinary bladder

Authors:

Alvaro Munoz (amunoz@bcm.edu)
David A Gangitano (dag006@shsu.edu)
Christopher P Smith (cps@bcm.edu)
Timothy B Boone (tboone@tmh.edu)
George T Somogyi (gsomogyi@bcm.edu)

Version: 5  Date: 12 May 2010

Author's response to reviews:

Dear Editor,

We have reviewed and changed the references to be cited in the text using consecutive numbers in square brackets. We are not using superscript numbers anymore. And reference 18 has now be cited in the correct order.

We checked our manuscript for any typographical errors and made final corrections.

Thanks.